Author’s response to reviews

Title: Burden and Risk Factors of Invasive Group B Streptococcus Disease among Neonates in a Chinese Maternity Hospital

Authors:
Qunhua Ying (ssbx1972@163.com)
Shutan Wang (526141016@qq.com)
Xiuming Lou (290465240@qq.com)
Jinlong Ding (41041924@qq.com)
Jiefeng Ding (djfyd@163.com)

Version: 1 Date: 18 Sep 2017

Author’s response to reviews:

Dear Caterer:

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript. We have revised our manuscript according to the comments:

1. Title Completed.

2. Declarations Revised.

3. Title page Revised.

4. Raw data

The datasets generated and analysed during the current study are not publicly available due to private protection but are available from the corresponding author on reasonable request.
We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,
Qunhua Ying